Author's response to reviews

Title: Validation of the Diffusion-weighted Imaging Method for Activity Evaluation in Crohn's Disease Patients Underwent Magnetic Resonance Enterography

Authors:

Qi Feng (fengqi_med@126.com)
Jun Shen (shenjun79@sina.cn)
Yuqi Qiao (medqyg@gmail.com)
Jiangchen Peng (pengjc0526@163.com)
Xiaochuan Geng (gengxiaochuan1949@126.com)
Jiong Zhu (billzhu49@hotmail.com)
Jianrong Xu (xujianrong@renji.com)

Version: 2
Date: 27 July 2014

Author's response to reviews:

We have added the line numbering and acknowledgement before you can send for peer-review.